Author's response to reviews

Title: Subject: Menstruation associated hypocalcemic symptoms and serum total calcium in patients with idiopathic hypoparathyroidism

Authors:

Soma Saha Dr (dr_somasaha5@rediffmail.com)
Ravinder Goswami Professor (gosravinder@hotmail.com)

Version: 2 Date: 19 December 2013

Author's response to reviews:

To
The BMC Endocrine Disorders Editorial Team
Thank you for your letter. As per the Instructions, we have now made necessary changes in the manuscript..
The changes made are as follows
1: Complete address of the authors including country and email
2: Background and conclusion in the manuscript body
3: Abbreviations
4. Competing statement
5. Authors’ contribution
6. Tables are inserted into the manuscript after references
Please do let us know if there are other requirements.
Thanking You
Yours sincerely
R Goswami